Author's response to reviews

Title: Nefopam for the prevention of perioperative shivering: A Meta-analysis of randomized controlled trials

Authors:

Meng Lv Mr (gylvmeng@163.com)
Xuetao Wang Mr (taoxuewang@vip.com)
Wendong Qu Mr (wendongqu@163.com)
Mengjie Liu MS (liumengjie31@163.com)
Yuelan Wang Professor (wyldgf@163.com)

Version: 2
Date: 8 March 2015

Author's response to reviews: see over
Dear editorial board of BMC anesthesiology,

Please find the revised manuscript: Nefopam for the prevention of perioperative shivering: a Meta analysis of randomized controlled trials, by Meng Lv. Et al., to be submitted as a research article to the BMC anesthesiology for consideration of publication. All co-authors have seen and agreed with the contents of the manuscript and there is no financial interest to report. We certify that the submission is original work and is not under review at any other publication.

In this manuscript, we have made changes of the reference list, first page of the manuscript. Besides, we included the line and page numbers in the manuscript. Also, a list of abbreviation was provided in this article.

Sincerely yours,

Meng Lv, on behalf of the authors,

Corresponding author: Yuelan Wang at Department of Anesthesiology, Qianfo Shan hospital, Shandong University, Jinan, Shandong, China.